

## GOPEN ACCESS

**Citation:** Cohen JL, Leslie HH, Saran I, Fink G (2020) Quality of clinical management of children diagnosed with malaria: A cross-sectional assessment in 9 sub-Saharan African countries between 2007–2018. PLoS Med 17(9): e1003254. https://doi.org/10.1371/journal.pmed.1003254

Academic Editor: Paul Garner, Liverpool School of Tropical Medicine, UNITED KINGDOM

Received: December 7, 2019

Accepted: August 10, 2020

Published: September 14, 2020

**Copyright:** © 2020 Cohen et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data used in this study cannot be shared publicly because they must be requested by the user. However, all of the data used in this study, other than the Ethiopia data, are available upon request at https://dhsprogram.com/ What-We-Do/Survey-Types/index.cfm. The Ethiopian SPA data can be requested at https:// ephi.gov.et/index.php/contact.

**Funding:** The authors received no specific funding for this work.

RESEARCH ARTICLE

# Quality of clinical management of children diagnosed with malaria: A cross-sectional assessment in 9 sub-Saharan African countries between 2007–2018

Jessica L. Cohen<sup>1\*</sup>, Hannah H. Leslie<sup>1</sup>, Indrani Saran<sup>2</sup>, Günther Fink<sup>3,4</sup>

1 Harvard T. H. Chan School of Public Health, Boston, Massachusetts, United States of America, 2 Boston College School of Social Work, Chestnut Hill, Massachusetts, United States of America, 3 Swiss Tropical and Public Health Institute, Basel, Switzerland, 4 University of Basel, Basel, Switzerland

\* cohenj@hsph.harvard.edu

## Abstract

## Background

Appropriate clinical management of malaria in children is critical for preventing progression to severe disease and for reducing the continued high burden of malaria mortality. This study aimed to assess the quality of care provided to children under 5 diagnosed with malaria across 9 sub-Saharan African countries.

## Methods and findings

We used data from the Service Provision Assessment (SPA) survey. SPAs are nationally representative facility surveys capturing quality of sick-child care, facility readiness, and provider and patient characteristics. The data set contained 24,756 direct clinical observations of outpatient sick-child visits across 9 countries, including Uganda (2007), Rwanda (2007), Namibia (2009), Kenya (2010), Malawi (2013), Senegal (2013–2017), Ethiopia (2014), Tanzania (2015), and Democratic Republic of the Congo (2018). We assessed the proportion of children with a malaria diagnosis who received a blood test diagnosis and an appropriate antimalarial. We used multilevel logistic regression to assess facility and provider and patient characteristics associated with these outcomes. Subgroup analyses with the 2013-2018 country surveys only were conducted for all outcomes. Children observed were on average 20.5 months old and were most commonly diagnosed with respiratory infection (47.7%), malaria (29.7%), and/or gastrointestinal infection (19.7%). Among the 7,340 children with a malaria diagnosis, 32.5% (95% CI: 30.3%-34.7%) received both a blood-testbased diagnosis and an appropriate antimalarial. The proportion of children with a blood test diagnosis and an appropriate antimalarial ranged from 3.4% to 57.1% across countries. In the more recent surveys (2013–2018), 40.7% (95% CI: 37.7%–43.6%) of children with a malaria diagnosis received both a blood test diagnosis and appropriate antimalarial. Roughly 20% of children diagnosed with malaria received no antimalarial at all, and nearly 10% received oral artemisinin monotherapy, which is not recommended because of

**Competing interests:** The authors have declared that no competing interests exist.

Abbreviations: ACT, Artemisinin Combination Therapy; RDT, rapid diagnostic test; SPA, Service Provision Assessment; WHO, World Health Organization. concerns regarding parasite resistance. Receipt of a blood test diagnosis and appropriate antimalarial was positively correlated with being seen at a facility with diagnostic equipment in stock (adjusted OR 3.67; 95% CI: 2.72–4.95) and, in the 2013–2018 subsample, with being seen at a facility with Artemisinin Combination Therapies (ACTs) in stock (adjusted OR 1.60; 95% CI:1.04–2.46). However, even if all children diagnosed with malaria were seen by a trained provider at a facility with diagnostics and medicines in stock, only a predicted 37.2% (95% CI: 34.2%–40.1%) would have received a blood test and appropriate antimalarial (44.4% for the 2013–2018 subsample). Study limitations include the lack of confirmed malaria test results for most survey years, the inability to distinguish between a diagnosis of uncomplicated or severe malaria, the absence of other relevant indicators of quality of care including dosing and examinations, and that only 9 countries were studied.

#### Conclusions

In this study, we found that a majority of children diagnosed with malaria across the 9 surveyed sub-Saharan African countries did not receive recommended care. Clinical management is positively correlated with the stocking of essential commodities and is somewhat improved in more recent years, but important quality gaps remain in the countries studied. Continued reductions in malaria mortality will require a bigger push toward quality improvements in clinical care.

#### Author summary

#### Why was this study done?

- Each year, millions of people become infected with malaria, and hundreds of thousands of people die from the disease, many of them young children in sub-Saharan Africa. Most deaths from malaria are avertable with timely provision of testing and treatment.
- Clinical guidelines for malaria include a blood test confirmation of disease prior to prompt treatment with effective antimalarials.
- Relatively little is known regarding provider compliance with these guidelines in malaria-endemic settings.

#### What did the researchers do and find?

- We used nationally representative clinical observations of outpatient sick-child care from 9 sub-Saharan African countries to assess clinical management of children with a malaria diagnosis.
- We find that 32.5% of children diagnosed with malaria had both a blood test diagnosis and received appropriate antimalarial treatment, with a wide range of 3.4% to 57.1% across countries.
- Health worker training and facility supplies of diagnostic tests and effective antimalarials do not appear to explain these gaps in quality of care.

 Clinical management is somewhat better for 2013–2018 surveys than in the earlier years because of higher rates of blood test diagnosis.

#### What do these findings mean?

- Despite increases in the distribution of malaria tests and effective antimalarial medications, significant gaps in the quality of care for pediatric malaria are present in these 9 countries.
- Further improvements in quality of malaria care may require a better understanding of remaining barriers and facilitators to appropriate management.

### Introduction

The year 2000 marked a turning point in global efforts to control malaria, which at the time was a leading cause of death among children under 5 in Africa [1]. Since then, malaria prevalence in sub-Saharan Africa has approximately halved; 19 countries globally have eliminated malaria altogether, and 20 more are on the verge of elimination [2,3]. This tremendous progress has been fueled by the widescale deployment of malaria prevention and treatment technologies that, together with unprecedented global malaria funding, have facilitated dramatic increases in the scale of malaria control efforts [3].

Despite this progress, the burden of malaria remains high, with over 200 million cases and an estimated 400,000 deaths per year [2]. Further, evidence suggests that progress has slowed in recent years, with some countries seeing increases in malaria transmission [2]. This has prompted the World Health Organization (WHO) and other global leaders in malaria control policy to call for a renewed focus on approaches to reducing the malaria burden. Further reductions in malaria morbidity and mortality will require new insights into gaps in the effectiveness of malaria control programs, including evidence on health system performance in the clinical management of pediatric malaria.

Most of the global malaria mortality burden is among young children in sub-Saharan Africa who are infected with the *Plasmodium falciparum* species of the parasite [2]. Children with *P. falciparum* malaria require prompt, appropriate antimalarial treatment in order to prevent progression of the disease to severe morbidity or death [4,5]. Since the symptoms of malaria overlap with a number of other common viral and bacterial diseases, blood test confirmation of the disease through microscopy or rapid test is strongly recommended [6,7]. WHO and national malaria control guidelines include blood test confirmation of malaria prior to treatment because testing can ensure better management of nonmalaria illness, reduce wasted resources on unnecessary antimalarial prescriptions, and improve health system surveillance data [6].

WHO and country-level guidelines for clinical management of malaria have evolved substantially in this century, including the introduction of Artemisinin Combination Therapies (ACTs) as first-line treatment for uncomplicated malaria between 2004–2006 and recommended testing prior to treatment for all ages between 2007–2012 [8]. Table 1 shows the recommended first-line treatment for uncomplicated and severe malaria for all countries in the sample, as well as the timing of guideline changes and the introduction of ACTs and rapid diagnostic tests (RDTs) [9–19]. Following these guideline changes, major efforts were made to improve access to testing and antimalarial treatment, including the free or heavily subsidized

| Survey  |   |  | Testing   |  |  |  |  |
|---|---|--|---|--|--|--|--|
| Country (survey<br>year)                      | N sick children (N<br>children with<br>malaria diagnosis) | First-line<br>treatment for<br>severe malaria <sup>1</sup> | First-line treatment for<br>uncomplicated<br>malaria <sup>1</sup> | Year ACTs became<br>first-line treatment<br>for malaria <sup>1</sup> | Year ACTs became<br>free/subsidized in<br>public sector <sup>1</sup> | Year testing before<br>treatment policy<br>for all ages <sup>1</sup> | Year<br>RDTs<br>rolled<br>out <sup>2</sup> |
| WHO <sup>3</sup>                              |   | AS   | AL, AS + AQ, AS<br>+ MQ, AS + SP,<br>DHA-PPQ                      | 2006   |  | 2010   |  |
| Democratic<br>Republic of the<br>Congo (2018) | 2,656 (2,027)   | AS, QN   | AS + AQ   | 2005   | 2006   | 2007   | 2011–<br>2012                              |
| Ethiopia (2014)                               | 1,898 (221)   | AS, AM, QN   | AL  | 2004   | 2004   | 2010   | 2010–<br>2011                              |
| Kenya (2010)                                  | 1,999 (1,049)   | AS, AM, QN   | AL  | 2004   | 2006   | 2009   | 2012                                       |
| Malawi (2013)                                 | 3,310 (954)   | AS, QN   | AL  | 2007   | 2007   | 2011   | 2011                                       |
| Namibia (2009)                                | 1,531 (110)   | QN   | AL  | 2006   | 2005   | 2012   | 2006                                       |
| Rwanda (2007)                                 | 1,662 (713)   | AS, QN   | AL  | 2005   | 2016   | 2009   | 2011–<br>2012                              |
| Senegal (2013–<br>2017)                       | 5,728 (102)   | AS, QN   | AL, AS + AQ,<br>Dha-ppq   | 2005   | 2010   | 2007   | 2007                                       |
| Tanzania (2015)                               | 4,950 (1,441)   | AS, AM, QN   | AL  | 2004   | 2006   | 2009   | 2009–<br>2010                              |
| Uganda (2007)                                 | 1,022 (723)   | AS, QN   | AL  | 2004   | 2006   | 2008   | 2011–<br>2012                              |

#### Table 1. Malaria policy changes and introduction of ACTs and RDTs by country.

<sup>1</sup>Source: World Malaria Report 2018, country profiles. https://www.who.int/malaria/publications/country-profiles/en/.

<sup>2</sup>Sources: [9-17].

<sup>3</sup>Sources: [<u>18,19</u>].

Abbreviations: AM, artemether; AL, artemether lumefantrine; AQ, amodiaquine; AS, artesunate; DHA-PPQ, dihydroartemisinin/piperaquine; MQ, mefloquine; NA, not applicable; QN, quinine; SP, sulfadoxine-pyrimethamine.

https://doi.org/10.1371/journal.pmed.1003254.t001

provision of ACTs to public sector health facilities and the large-scale distribution of malaria RDTs (Table 1). Evidence suggests that the scale-up of RDTs has contributed to higher rates of malaria testing [2,9,20] and that the widescale distribution of subsidized ACTs has contributed to declines in malaria morbidity and mortality [3,21], yet important gaps in clinical management of malaria remain [22–25].

Several health system challenges are likely contributing to inadequate clinical management of the disease and thus to the enduring malaria mortality burden among young children [26]. First, caregivers often do not seek formal facility-based care for children with malaria symptoms, instead using over-the-counter medications, informal care, or no care at all [27–29]. Second, even children who are seen by a formal provider may not receive appropriate care [28]. Although empirical evidence on the role of provider quality in pediatric malaria deaths is limited [30–32], a recent study estimates that up to 50% of deaths from infectious diseases in low-income settings can be attributed to patients receiving low-quality care [33].

While a number of studies have assessed quality of care for malaria patients, most are based on interviews with patients or record reviews, though some studies at the national or subnational level also include direct observation of clinical practice [22,24,34–36]. One cross-country study that explores management of febrile illnesses using observation with a standardized checklist was conducted in the context of a program to improve clinical care with supportive supervision [37]. To our knowledge, no studies have used a uniform tool for clinical observation of malaria management across a range of countries to assess the quality of care for malaria

under usual settings. This study uses clinical observation data from nearly 25,000 sick-child visits across 6,400 health facilities in 9 sub-Saharan African countries and spanning a 12-year period to evaluate the extent to which children with a malaria diagnosis received a blood test diagnosis and appropriate antimalarial treatment. We report estimates of these critical process measures of quality of care and explore potential predictors of quality, including facility stocking of essential malaria commodities and health worker training.

#### Methods

#### Data

The data used in this study are from the Service Provision Assessment (SPA) survey, implemented by ICF International [38]. All of the questionnaires can be found on the SPA website (https://dhsprogram.com/What-We-Do/Survey-Types/SPA-Questionnaires.cfm). We use data from 4 modules in the SPA survey battery: 1) direct observations of sick-child consultations for children under age 5, 2) facility surveys assessing service readiness, 3) interviews with healthcare providers, and 4) exit interviews with caregivers of the sick children observed in the consultation survey. All SPA surveys conducted in Africa in 2007 and later were analyzed, including the Democratic Republic of the Congo (2018), Ethiopia (2014), Kenya (2010), Malawi (2013), Namibia (2009), Rwanda (2007), Senegal (2013-2017), Tanzania (2015), and Uganda (2007). SPA surveys in Malawi and Namibia were a census of all registered public and private facilities in the country, whereas Rwanda was a full census of all public facilities and all large private facilities. In all other countries, the SPA survey was based on a nationally representative facility sample drawn from a master list and stratified by region and public/private ownership. All district-level and higher-level hospitals were included in the survey, and lowerlevel facilities were selected with a preset probability, with the lowest sampling fraction for the lowest tier facilities. In Senegal, health huts were administered a shorter assessment without observations of care and are therefore not included in the analysis. The SPA in Senegal was conducted annually for 5 years; we pool results across these years when presenting countryspecific estimates. SPA surveys conducted before 2007 were excluded because they lacked sufficient detail on the type of antimalarial treatments prescribed.

The SPA surveys include direct observations of outpatient consultations with sick children under age 5. Sick children present on the assessment day were listed and sampled at a systematic interval after a random start number with the goal of sampling up to 5 observations per provider (maximum of 15 per facility). Direct observations were conducted by trained fieldworkers with clinical experience (typically nurses). Following the observation, providers reported to the interviewer their diagnosis and treatment plan. One of the diagnoses recorded in the SPA was a malaria diagnosis. Observers recorded whether the provider reported a malaria diagnosis (based on either RDT or microscopy) and also whether the provider reported that the malaria diagnosis was reached based on symptoms or based on a blood test. They also recorded which medications (if any) were prescribed or directly given to the child. SPA guidelines dictated that enumerators follow patients through their entire visit whenever possible [39]. An interview with the child's caregiver was conducted after the observation for basic information on demographic characteristics, child symptoms, etc.

A provider interview and facility assessment were conducted on the same day as the clinical observation. We use information from the provider interview on the cadre and training in malaria diagnosis or treatment. From the facility assessment, we follow [40] and consider a facility to be equipped with malaria treatment and diagnosis equipment if the facility has observed, nonexpired artemether lumefantrine or artesunate-amodiaquine (the most common types of ACTs) in stock (any pack type) in the pharmacy area and had either observed,

nonexpired RDTs in the laboratory or malaria service area and/or an observed microscope, blood slides, and staining equipment in the laboratory. These measures of observed, nonexpired equipment should be highly—but not perfectly—correlated with the availability of these commodities at the facility since it is possible that commodities were available in other locations (such as the child curative care area if separate from the malaria service area) or that providers are using expired medications or dosages.

A total of 25,426 sick-child observations were available in the SPA surveys. 274 observations were dropped because the child was 60 months or older. We also dropped 391 observations in which the SPA-supplied patient weight was missing or 0 (which indicates that either the clinical observation or caregiver interview were not consented to or not completed) and 5 observations in which the malaria diagnosis variable was missing.

#### Analysis

The analysis was not based on a prespecified analysis plan. Observations were eligible for analysis if the provider reported diagnosing malaria, regardless of whether any other, additional, diagnoses were made (for example, respiratory infection). This inclusion criterion was used because SPA surveys only routinely inquire about malaria blood testing for those children ultimately diagnosed with malaria (not for children who had exclusively a nonmalaria diagnosis). WHO and country guidelines call for diagnostic confirmation of malaria by blood test and treatment with an ACT for uncomplicated malaria or parenteral artemisinin or quinine for severe malaria (Table 1) [6]. We assess the extent to which children with a malaria diagnosis receive 1) a blood-test–based diagnosis, 2) an appropriate antimalarial, and 3) both a blood test diagnosis and appropriate antimalarial.

The definition of these outcomes is restricted by several drawbacks in the SPA survey tool. First, for most SPA surveys, malaria test results are not reported—the SPA only includes a variable indicating whether or not a malaria diagnosis was based on a blood test. Specifically, for all children diagnosed with malaria, observers note whether the provider reports basing this decision on a blood test (either an RDT or microscopy) or based on symptoms only. While only available for the Tanzania (2015) and DRC (2018) surveys, observers in these surveys also checked malaria test results when a blood test diagnosis was reported and did confirm that 91.4% of these children had a record of a positive test. Second, the SPA surveys do not distinguish between diagnoses of severe or uncomplicated malaria. Accordingly, we considered a child to be treated with an "appropriate antimalarial" if they received an appropriate treatment for either uncomplicated or severe malaria, including oral ACTs, parenteral artemisinin, or injectable quinine. To the extent that children with uncomplicated malaria received treatment for severe malaria (or vice versa), this definition of "appropriate antimalarial" would be generous (an upper bound). The third limitation of the SPA surveys is that they do not report on the dosage of medication given, so we cannot determine whether the dose of antimalarial given was appropriate, again leading to the "appropriate antimalarial" definition being an upper bound of appropriate prescribing.

On the other hand, the SPA surveys do have detailed information about the types of antimalarials that were prescribed. We computed the fraction of children with a malaria diagnosis who received no antimalarial, an oral ACT, parenteral artemisinin, injectable quinine, oral artemisinin monotherapy, or another oral antimalarial (principally, chloroquine, amodiaquine, sulfadoxine-pyremethamine, and quinine). Oral artemisinin monotherapy is not recommended under any circumstances because of concerns about parasite resistance [41], and the "other antimalarials" are no longer recommended because of decreased efficacy [6]. We computed the fraction of children diagnosed with malaria who received each of these prescriptions as well as combinations of prescriptions for all combinations that included at least 10 cases. This excluded 24 malaria cases (0.25%).

We then assessed factors associated with receipt of both a blood test diagnosis and an appropriate antimalarial. We estimated a multilevel logistic regression model with a binary outcome variable equal to 1 if the child received both a blood test diagnosis and an appropriate antimalarial on a set of patient, caregiver, provider, and facility characteristics as well as survey year fixed effects. This analysis excludes an additional 385 observations (5%) because of missing covariates. We used cluster-robust standard errors to account for repeated measurements within facilities [42]. We use these adjusted models to predict receipt of both blood test diagnosis and appropriate antimalarial under the hypothetical scenario of all facilities having observed, valid ACTs and malaria testing equipment in stock, as well as all providers reporting training in malaria diagnosis and/or treatment.

Given the time it takes for countries to adapt to clinical guideline changes and the fact that ACTs and RDTs have become increasingly available in later years, the emphasis on testing and appropriate treatment (and access to tests and antimalarials) may be higher in recent years. Thus, for all outcomes, in addition to the pooled survey results, we report outcomes for the subset of country surveys conducted in 2013 and later (Malawi [2013], Senegal [2013–2017], Ethiopia [2014], Tanzania [2015], and Democratic Republic of the Congo [2018]). The 9 countries also have a range of health system structures and epidemiological profiles, so we perform a "leave one out" sensitivity analysis in <u>S3 Table</u>, in which we report results when dropping 1 country at a time from the data set.

All analyses were conducted using SPA-provided sampling weights, rescaled to maintain a weighted sample size equal to the observed sample size. The resulting estimates can be considered representative of all sick children under 5 seeking formal facility-based care in the countries included in our analysis at the time of the survey. All analyses were done using Stata Version 14 [43].

#### **Ethics statement**

This analysis of secondary data was deemed exempt from human subject review by the Harvard Office of Human Research Administration (IRB15-3668), and no funding was received for this study. The study team did not seek informed consent from survey participants because the secondary data used in the study were anonymous, but the SPA surveyors obtain written consent from facility managers, providers, and patients when collecting the data.

#### Results

A total of 24,756 sick-child clinical observations across 6,453 facilities were included in the pooled analysis (18,542 observations and 4,883 facilities within the 5-country subset from 2013–2018). Of all the facilities, 6.8% were hospitals and 70.6% were health centers (S1 Table). Most of the remaining facilities were dispensaries and lower-level clinics. 48.8% of facilities had at least one child with a malaria diagnosis in the sample. Among these facilities, 65.7% had observed and valid ACTs and malaria testing equipment in stock (79.7% in the 2013–2018 sub-sample) (S1 Table).

Sample characteristics for the pooled sample are presented in Table 2 and for the 2013–2018 subset in S2 Table. Overall, 7,340 children (29.6%) were diagnosed with malaria (25.6% in the 2013–2018 subset). Children diagnosed with malaria were, on average, 23.2 months old; the average age for the full sample was 20.5 months. Multiple diagnoses were common, with 42% and 15.6% of the children diagnosed with malaria also diagnosed with a respiratory infection or a gastrointestinal infection, respectively. Most malaria cases (85%) were treated outside

#### Table 2. Sample characteristics.

|   | Children with Malaria Diagnosis<br>N = 7,340 | Children without Malaria Diagnosis<br>N = 17,416 | Valid N |
|---|--|--|---------|
|   | N (%)  | N (%)  |         |
| Child characteristics   |  |  |         |
| Age of child (months, mean ± SD)  | 23.2 ± 15.3                                  | $19.4 \pm 15.4$                                  | 24,382  |
| Child is female   | 3,589 (49%)                                  | 8,248 (47.4%)                                    | 24,704  |
| Child diagnosis   |  |  |         |
| Malaria   | 7,340 (100%)                                 | 0 (0%)   | 24,756  |
| Respiratory infection   | 2,997 (42%)                                  | 8,031 (51.1%)                                    | 22,858  |
| Gastrointestinal infection  | 1,111 (15.6%)                                | 3,337 (21.3%)                                    | 22,858  |
| Caregiver characteristics   |  |  |         |
| Caregiver age (years, mean ± SD)  | $28.3 \pm 8.1$                               | $28.4 \pm 8.4$                                   | 23,581  |
| Caregiver, primary education  | 3,764 (51.3%)                                | 7,266 (41.7%)                                    | 24,756  |
| Caregiver, some secondary education                                       | 1,896 (25.8%)                                | 4,651 (26.7%)                                    | 24,756  |
| Facility ownership  |  |  |         |
| Private facility  | 2,002 (27.3%)                                | 3,283 (18.8%)                                    | 24,756  |
| Facility level  |  |  |         |
| Hospital  | 1,101 (15%)                                  | 2,970 (17.1%)                                    | 24,756  |
| Health center   | 4,390 (59.8%)                                | 11,388 (65.4%)                                   | 24,756  |
| Other (health post, dispensary, etc.)                                     | 1,849 (25.2%)                                | 3,057 (17.6%)                                    | 24,756  |
| Facility stocking   |  |  |         |
| Has observed/verified malaria testing equipment                           | 5,514 (75.1%)                                | 14,879(85.4%)                                    | 24,756  |
| Has observed/verified appropriate antimalarial (ACT) in stock             | 6,524 (88.9%)                                | 14,985 (86%)                                     | 24,756  |
| Has both appropriate malaria testing equipment and antimalarial treatment | 4,985 (67.9%)                                | 13,137 (75.4%)                                   | 24,756  |
| Provider characteristics  |  |  |         |
| MD or MO  | 509 (7%)                                     | 1,542 (8.9%)                                     | 24,476  |
| Paramedical (for example, clinical officer, advanced practice clinician)  | 2,395 (33%)                                  | 6,869 (39.8%)                                    | 24,476  |
| Nurse or other provider type (for example, CHW, aide)                     | 4,364 (60%)                                  | 8,832 (51.2%)                                    | 24,476  |
| Provider trained in malaria diagnosis or treatment                        | 4,433 (62.9%)                                | 10,785 (69.5%)                                   | 22,541  |

All estimates are weighted to be nationally representative of sick children under 5 seeking facility-based care in the year of the survey. **Abbreviations**: ACT, Artemisinin Combination Therapy; CHW, community health worker; MO, medical officer.

https://doi.org/10.1371/journal.pmed.1003254.t002

of hospitals, in primary care health centers or smaller health posts, and only 7% were seen by a medical doctor or medical officer. 62.9% of malaria cases were seen by a provider with training in malaria diagnosis and/or treatment (62.6% in the 2013–2018 subset), and 67.9% were seen at a facility with observed malaria testing equipment and appropriate antimalarials in stock (80.3% in the 2013–2018 subset).

Clinical management of children diagnosed with malaria is presented in Fig 1 and S3 Table. Among children diagnosed with malaria, 56.4% (95% CI: 53.9%–58.8%) received a blood test diagnosis overall, with a range of 12.5% (95% CI: 4.9%–20.1%) in Namibia to 80.9% (95% CI: 72.2%–89.6%) in Ethiopia. Among all children diagnosed with malaria, 58.8% (95% CI: 56.5%–61.2%) were prescribed an appropriate antimalarial, with prescribing ranging from 18.2% (95% CI: 9.2%–27.2%) in Ethiopia to 80.1% in Uganda (95% CI: 74.8%–85.4%). Overall, 32.5% (95% CI: 30.3%–34.7%) of children with a malaria diagnosis received both a blood test diagnosis and an appropriate antimalarial, ranging from 3.4% (95% CI: 0.3%–6.5%) in Namibia to 57.1% (95% CI: 51.6%–62.5%) in Malawi. When restricted to children seen at





https://doi.org/10.1371/journal.pmed.1003254.g001

hospitals, 36.4% (95% CI: 31.1%–41.2%) received both a blood test diagnosis and appropriate antimalarial (S3 Table). These estimates are not very sensitive to dropping any particular country–year (S3 Table).

Blood test diagnoses appear to be more common in the later country surveys, although prescription of an appropriate antimalarial does not. Receipt of both a blood test diagnosis and appropriate antimalarial is somewhat higher in the 2013–2018 subset than in the pooled analysis, with 40.7% (95% CI: 37.7%–43.6%) of children diagnosed with malaria receiving both in these years (S3 Table).

Turning to specific prescribing patterns, 20.1% (95% CI: 19.2%–21%) of the children diagnosed with malaria were not prescribed any antimalarial (Fig 2; country-specific results in S4 Table). 50.4% (95% CI: 49.3%–51.6%) were prescribed an ACT (48.1% prescribed an ACT alone and 2.3% in combination with other medications), and less than 0.5% were prescribed parenteral artemisinin. 9.7% (95% CI: 9.0% to 10.4%) of children with a malaria diagnosis were prescribed a quinine injection (6.0% alone and 3.7% in combination with other medications). Oral artemisinin monotherapy was prescribed to 9.7% (95% CI: 9.1%–10.4%) of children with a malaria diagnosis (9.2% alone and 0.5% in combination with other medications). Finally, 13.6% (95% CI: 12.8%–14.4%) of children with a malaria diagnosis were prescribed other, older antimalarials (11.8% alone and 1.8% in combination with other medications). Antimalarial prescribing patterns were very similar for the 2013–2018 subset as for the pooled analysis (Fig 2 and S4 Table), with 47.9% receiving an oral ACT, 16.9% receiving no antimalarial, and 16% receiving older antimalarials.





https://doi.org/10.1371/journal.pmed.1003254.g002

Results from multivariable logistic regressions are presented in Table 3 (crude ORs are in S5 Table). The odds of receiving a blood test diagnosis and an appropriate antimalarial were higher for older children than for infants. Children diagnosed with only malaria had 1.33 (95% CI: 1.09–1.61) higher odds of receiving a blood test diagnosis and an appropriate antimalarial than children diagnosed with both malaria and another illness. The odds of receiving a blood test diagnosis and an appropriate antimalarial were 3.67 (95% CI: 2.72-4.95) times higher for children being seen at facilities that had observed and valid malaria testing equipment in stock than for those who were not. This association was even stronger (OR 4.34; 95% CI: 2.66-7.10) for the 2013–2018 subsample. The association between being seen at a facility that had observed and valid ACTs in stock and receiving a blood test diagnosis and appropriate antimalarial was positive, but insignificant, in the full sample (1.31; 95% CI: 0.91-1.90) but somewhat stronger and significant in the 2013-2018 subsample (1.60; 95% CI: 1.04-2.46). No significant association was found between being seen by a provider with training in malaria diagnosis or treatment and receipt of a blood test and appropriate antimalarial. The odds of receiving both a blood test diagnosis and an appropriate antimalarial were significantly lower for children seen by a medical doctor or medical officer than children seen by a nurse but were higher for children seen by a paramedical provider (such as a clinical officer) than by a nurse in both the full sample and the 2013-2018 subsample.

Estimates from the logistic regressions presented in Table 3 imply that, if all children with a malaria diagnosis were seen at a facility that had observed ACTs and diagnostic testing equipment in stock and were seen by a provider with training in malaria management, a predicted 37.2% (95% CI: 34.3%–40.2%) of them would receive both a blood test diagnosis and an appropriate antimalarial in the pooled analysis (Fig 3). For the 2013–2018 subset, this predicted

|  | All Years (N = 6,963) |               |         | 2013-2018 Only (N = 4,496) |               |         |
|--|-----------------------|---------------|---------|----------------------------|---------------|---------|
|  | Adjusted OR           | 95% CI        | p-value | Adjusted OR                | 95% CI        | p-value |
| Child characteristics  |                       |               | -       |                            |               | -       |
| Age 12–23 months (ref: 0–11 months)  | 1.43                  | 1.17–<br>1.74 | 0.00    | 1.48                       | 1.16–<br>1.88 | 0.00    |
| Age 24–35 months (ref: 0–11 months)  | 1.53                  | 1.21-<br>1.93 | 0.00    | 1.47                       | 1.11–<br>1.94 | 0.01    |
| Age 36–47 months (ref: 0–11 months)  | 1.36                  | 1.05-<br>1.76 | 0.02    | 1.47                       | 1.07-<br>2.01 | 0.02    |
| Age 48–60 months (ref: 0–11 months)  | 1.41                  | 1.07-<br>1.86 | 0.02    | 1.46                       | 1.05-<br>2.04 | 0.03    |
| Female (ref: male)   | 0.98                  | 0.84-<br>1.13 | 0.76    | 1.02                       | 0.85-<br>1.22 | 0.83    |
| Child diagnosis  |                       |               |         |                            |               |         |
| Malaria only (ref: malaria + other illness)  | 1.33                  | 1.10–<br>1.61 | 0.00    | 1.32                       | 1.07–<br>1.64 | 0.01    |
| Caregiver characteristics  |                       |               |         |                            |               |         |
| Primary education (ref: no education)  | 0.94                  | 0.77-<br>1.15 | 0.54    | 0.97                       | 0.76-<br>1.23 | 0.78    |
| Some secondary education (ref: no education)   | 0.78                  | 0.60-<br>1.02 | 0.07    | 0.71                       | 0.53–<br>0.96 | 0.02    |
| Facility ownership   |                       |               |         |                            |               |         |
| Private facility (ref: public facility)  | 0.88                  | 0.69–<br>1.12 | 0.30    | 0.69                       | 0.52-<br>0.92 | 0.01    |
| Facility level   |                       |               |         |                            |               |         |
| Hospital (ref: health post, dispensary)  | 1.1                   | 0.75–<br>1.61 | 0.62    | 1.10                       | 0.70-<br>1.74 | 0.67    |
| Health center (ref: health post, dispensary)   | 1.15                  | 0.81-<br>1.63 | 0.45    | 1.18                       | 0.79–<br>1.77 | 0.42    |
| Facility stocking  |                       |               |         |                            |               |         |
| Has valid/verified ACT in stock (ref: no valid/verified ACT in stock)  | 1.31                  | 0.91–<br>1.90 | 0.15    | 1.60                       | 1.04–<br>2.46 | 0.03    |
| Has valid/verified malaria testing equipment (ref: no valid/verified malaria testing equipment)                    | 3.67                  | 2.72-<br>4.95 | 0.00    | 4.34                       | 2.66–<br>7.10 | 0.00    |
| Provider characteristics   |                       |               |         |                            |               |         |
| MD or MO (ref: nurse or other provider type)   | 0.47                  | 0.33-<br>0.68 | 0.00    | 0.52                       | 0.35-<br>0.76 | 0.00    |
| Paramedical (for example, clinical officer, advanced practice clinician) (ref: nurse or other provider type)       | 1.45                  | 1.04-<br>2.03 | 0.03    | 1.61                       | 1.05-<br>2.46 | 0.03    |
| Provider trained in malaria diagnosis or treatment (ref: provider never trained on malaria diagnosis or treatment) | 0.85                  | 0.69–<br>1.06 | 0.15    | 0.85                       | 0.66-<br>1.09 | 0.21    |

#### Table 3. Patient, provider, and facility correlates with receipt of blood test diagnosis and recommended medication for malaria (logistic regression).

Outcome variable is binary variable for diagnosis based on blood test and receipt of appropriate antimalarial. Coefficients are ORs from logistic regressions including all variables presented in table and survey year fixed effects. Standard errors are adjusted for clustering within facilities, and data are weighted using SPA-supplied sampling weights. **Abbreviations**: ACT, Artemisinin Combination Therapy; MO, medical officer; SPA, Service Provision Assessment.

https://doi.org/10.1371/journal.pmed.1003254.t003

estimate increases modestly to 44.4% (95% CI: 40.6%–48.1%) of children with a malaria diagnosis receiving both a blood test diagnosis and appropriate medication. The difference between the observed and predicted levels is largest for receipt of a test and for receipt of both a test and appropriate medication, reflecting the strong association between test availability and testing shown in Table 3.



**Fig 3. Estimated prevalence of clinical management of children diagnosed with malaria with universal stocking and provider training (N = 6,955 child visits).** Estimates are among the sample of children diagnosed with malaria for which covariates included in the model presented in Table 3 are available. "Blood test diagnosis" indicates that the child's malaria diagnosis was based on either a blood slide microscopy test or a rapid diagnostic test. "Appropriate medication" indicates that the child received an oral ACT, parenteral artemisinin, or injectable quinine. Predicted estimates are based on the model presented in Table 3, under the assumption that all children are seen at a facility that had observed, verified stocking of malaria diagnostic tests and ACTs and that all children are seen by a provider with training in malaria diagnosis and/or treatment, holding all other covariates at their mean value. Data are weighted by SPA-supplied sampling weights to be nationally representative, and 95% confidence intervals are adjusted for clustering within facilities. ACT, Artemisinin Combination Therapy; SPA, Service Provision Assessment.

https://doi.org/10.1371/journal.pmed.1003254.g003

### Discussion

This paper uses a large clinical observation data set on malaria management to assess the quality of clinical care for children diagnosed with malaria in 9 sub-Saharan African countries. We find wide variation across countries in receipt of a blood test diagnosis and an appropriate antimalarial prescription. Overall, we find that two-thirds of children with a malaria diagnosis either did not receive a blood test diagnosis or were not given a recommended antimalarial both essential process measures of quality of malaria care. While clinical management appears somewhat better in the 2013–2018 period—for diagnostic testing in particular—still, only about 40% of children with a malaria diagnosis are receiving both a blood test diagnosis and appropriate medication in these years, and 1 in 5 children with a malaria diagnosis do not get any antimalarial at all.

Diagnostic confirmation of malaria is a central recommendation of malaria treatment guidelines. Symptom-based diagnosis often yields high rates of antimalarial treatment for non-malarial illnesses, which can lead to inappropriate ACT prescribing, wasted subsidy dollars, and increased likelihood of parasite resistance [44,45]. Nearly half of children with a malaria diagnosis in this study did not receive a diagnosis based on a blood test. The malaria diagnostic capacity of facilities in the countries surveyed was high, but not ideal, with over 20% of facilities lacking functioning diagnostic equipment. Being seen at a facility with observed diagnostic equipment was highly correlated with quality of care, suggesting that increased emphasis on

testing and distribution of testing equipment is important for continued progress in malaria management. Diagnostic testing was also more common in the later survey years (2013–2018), which likely reflects the fact that RDTs were rolled out in the sample countries between 2010–2012 (Table 1) and that the emphasis on testing grew following the 2010 WHO guideline change to encourage parasitological confirmation of malaria prior to antimalarial treatment for all ages.

Treatment with ACTs is critical to preventing the progression of malaria to severe disease [4]; untreated severe malaria has an estimated case fatality rate of 13%–21% [5]. We find that roughly one-third of children diagnosed with malaria were either prescribed no antimalarial at all or were prescribed an antimalarial that is not recommended due to reduced efficacy such as chloroquine or sulfadoxine-pyrimethamine. Although parenteral artesunate is superior to injectable quinine for severe malaria [46], prescriptions for quinine injections were much more common than for parenteral artesunate. This likely reflects the limited availability of parenteral artesunate because in 2011, only 1 source had been prequalified by WHO [47]. Finally, we find that nearly 10% of children diagnosed with malaria were prescribed artemisinin monotherapy, which is strongly discouraged by WHO because of concerns about worsening parasite resistance [6,48]. Starting in 2007, WHO instructed member countries to begin phasing out monotherapy [41], so some prescribing of artemisinin monotherapy in the early survey years could be due to progressive phase-out. However, prescribing of artemisinin monotherapy persisted in the later surveys as well (S4 Table).

Stocking of ACTs was incomplete, with 85% of facilities having observed, valid ACTs in stock. Although ACTs became WHO's recommended first-line treatment for malaria in 2006 and most of the study countries began making ACTs free or heavily subsidized in the public sector around the same time (Table 1), appropriate antimalarial prescribing was not different for the 2013–2018 subgroup than for the pooled estimates. Being seen at a facility with ACTs was positively associated with malaria management—although not significantly so in the full sample—suggesting that consistent stocking of ACTs is likely an enduring, important consideration in the quality of malaria management.

There are a number of reasons why healthcare workers may not be providing blood tests and appropriate antimalarials to children with a malaria diagnosis in the study countries. Testing may be incomplete because facilities lack a licensed microscopist or lab technician, because providers do not trust the tests or feel that their clinical judgment is superior, or because of high patient volumes and/or high out-of-pocket costs of the tests [34,35,49]. Health workers may not be prescribing appropriate antimalarials because the appropriate dosage or suspension is not available, because of provider or patient preference for alternative medications, or because of high out-of-pocket costs for these medications [49–51]. Some studies have found that patients continue to face out-of-pocket costs for health services and commodities that are intended to be provided free in settings similar to the study countries [52–54]. Continued improvements in the quality of care for malaria will require a deeper investigation into the lingering barriers and facilitators of appropriate clinical management of the disease.

While the availability of malaria tests and antimalarial medication was correlated with clinical management of malaria, provider training in malaria was not. While some previous research finds an association between health worker training and quality of care for malaria [55], training interventions have generally had small effects on provider malaria management [56–58] and on adherence to clinical guidelines more generally [59,60]. There are many factors beyond training that can influence provider adherence to guidelines, such as weak incentives and accountability, insufficient preservice education, high patient/provider ratios, or other health system factors [61–64]. Our study did not include a detailed measure of provider knowl-edge and competence, nor did it capture potentially important factors related to provider

incentives, stress, and burnout. Estimates of these aspects of provider motivation alongside robust measures of quality of care are priorities for future research seeking to improve clinical management of malaria.

Our results are consistent with prior evidence from population-representative household surveys suggesting important gaps in diagnostic testing and ACT taking for children with malaria [22,25,65]. While these studies have the advantage of capturing children who are not brought for formal care, household surveys may be subject to recall bias and misunderstanding of clinical actions [66,67]. Our results underscore concerns about insufficient treatment for malaria and highlight the fact that even children who are brought to formal providers often receive inadequate care. Our results are also consistent with previous research on provider management of malaria, which has found that providers often fail to comply with testing and treatment guidelines but that compliance is somewhat responsive to guideline changes and the availability of equipment [9,14,68–75].

Our study was limited by the way in which the SPA survey records malaria testing, which is to indicate whether a child with a malaria diagnosis had a blood-test-based diagnosis or a clinical diagnosis. Because test results are not directly captured in SPA data, it is possible that the diagnosis may in some cases diverge from the test results. A number of previous studies have highlighted insufficient provider adherence to malaria test results [34,35,76]. While only available for the Tanzania (2015) and Democratic Republic of the Congo (2018) surveys, observers in these surveys checked malaria test results when tests were reported and did confirm that 91.4% of children diagnosed with malaria by blood test had a record of a positive test. Although it is possible that providers or lab technicians could have misrecorded test results in these settings, this is generally supportive that most children for whom the provider reported a blood-test-based diagnosis of malaria actually had a positive test result. The SPA also does not allow us to explore malaria testing for children who did not receive a malaria diagnosis (or whose malaria diagnosis was reported to be based on clinical symptoms), so we are unable to explore the important aspect of quality of care related to blood testing for all children with fever. Our assessment of quality of care for children diagnosed with malaria is, therefore, a narrow assessment of quality. Broader evaluations would include examinations and tests performed and their results, as well as provider counseling and communication.

Clinical observation data can be subject to Hawthorne effects and observer error [77,78]. For example, providers prescribing ACTs may have referred to them as "artemisinin," resulting in misclassification of ACTs as oral artemisinin monotherapy. In the 2015 Tanzania SPA survey, observers confirmed 94.3% of observer-recorded ACT prescriptions and 93.7% of oral artemisinin monotherapy prescriptions using medical records. Another limitation of the SPA is that it reports on what medicines are prescribed, not necessarily which medicines are obtained or consumed. The SPA does, however, report whether the patient leaves the facility with medications in hand. 92% of children diagnosed with malaria are reported to leave the facility with at least one medication, although we cannot confirm that these are the same as those prescribed.

Patient flow through facilities could have influenced the quality of the observation data for example, if enumerators were unable to follow patients from the provider exam to a lab for testing. SPA guidelines dictated that enumerators follow patients through their entire visit whenever possible [39]. Furthermore, in the surveys conducted after 2012, caregivers were asked in the exit interview whether they were sent to a lab within the facility or another provider for testing, and—among caregivers of patients who were recorded as having been diagnosed with malaria by blood test—78% of caregivers reported that they were sent for testing. This is likely a lower bound on the fraction of patients who actually were tested since it does not capture cases in which patients were tested by the primary provider within the exam room (for example, by RDT).

The countries in sub-Saharan Africa in which SPA conducted observations of care include more countries in East Africa than other parts of the subcontinent, and our results should be interpreted with this in mind. Furthermore, the countries included in the analysis have different epidemiological and health system profiles and may have been conducted in different seasons, which can have important implications for malaria prevalence and treatment. Although malaria epidemiology and management differ across the study countries, the overall results are not sensitive to the removal of specific surveys from the data set, suggesting that no particular country is driving the results. The results presented here are directly applicable to the 9 countries only. However, the consistent deficits in quality of care across these quite different countries suggests many other countries with high burden may experience similar challenges. High-quality evidence on the quality of malaria management—such as that based on observations of care—is urgently needed across a broader range of malaria-endemic countries, particularly those with the highest malaria burden that are the focus of the recent WHO "high burden to high impact" malaria strategy [79].

Two additional limitations to the study's findings are the inability to draw firm conclusions regarding temporal changes in malaria management and the fact that regression estimates reflect correlations between patient-, provider-, and facility-level factors and clinical management, not necessarily causal effects. The estimates of predicted malaria management under the hypothetical scenario of all facilities having test equipment and antimalarials in stock are informative but could be biased by unobserved confounding factors related to both stocking and clinical management. The malaria landscape has evolved over the study period, but it is inappropriate to compare estimates of malaria management from earlier years to later years and make inference about changes over time since the surveys conducted in different years were also conducted in different countries. It is reasonable to infer from the comparison of the pooled estimates to those from 2013-2018 that testing has improved and that clinical management has improved modestly, but, again, any inference about changes over time is confounded by the changing sample composition. Given frequently changing guidelines, technological innovations, and investments in diseases such as malaria, integrating routine collection of healthcare quality data into health information systems is a key priority for quality monitoring and improvement [80].

#### Conclusions

After many years of declining morbidity and mortality, malaria is at a crossroads [81]. Global funding for malaria programs has plateaued, and declines in malaria infection have stalled or possibly reversed. The results presented in this paper suggest that important gaps remain in the clinical management of malaria in the surveyed countries. Major efforts in quality of care will be needed to reduce the burden of malaria in the coming years.

#### Supporting information

**S1 Table. Facility characteristics.** (DOCX)

**S2** Table. Sample characteristics for 2013–2018 surveys only. (DOCX)

S3 Table. Clinical management of children diagnosed with malaria: Estimates by time period, country, hospital, and leaving one country out at a time. (DOCX)

**S4** Table. Antimalarial prescriptions among children diagnosed with malaria. (DOCX)

S5 Table. Patient, provider, and facility correlates with receipt of blood test diagnosis and recommended medication for malaria (logistic regression)—crude ORs. (DOCX)

#### Acknowledgments

We thank Naveen Sunder for his assistance with revisions of tables and figures.

### **Author Contributions**

Conceptualization: Jessica L. Cohen, Günther Fink.

Data curation: Jessica L. Cohen.

Formal analysis: Jessica L. Cohen, Hannah H. Leslie, Indrani Saran.

Investigation: Jessica L. Cohen.

Methodology: Jessica L. Cohen.

Visualization: Hannah H. Leslie, Indrani Saran.

Writing - original draft: Jessica L. Cohen.

Writing - review & editing: Jessica L. Cohen, Hannah H. Leslie, Indrani Saran, Günther Fink.

#### References

- 1. Bryce J, Boschi-Pinto C, Shibuya K, Black RE. WHO estimates of the causes of death in children. The Lancet. 2005; 365: 1147–1152. https://doi.org/10.1016/S0140-6736(05)71877-8
- 2. World Health Organization. World malaria report 2019. Geneva, Switzerland: World Health Organization; 2019.
- Bhatt S, Weiss DJ, Cameron E, Bisanzio D, Mappin B, Dalrymple U, et al. The effect of malaria control on Plasmodium falciparum in Africa between 2000 and 2015. Nature. 2015; 526: 207–211. <u>https://doi.org/10.1038/nature15535</u> PMID: 26375008
- Sinclair D, Zani B, Donegan S, Olliaro P, Garner P. Artemisinin-based combination therapy for treating uncomplicated malaria. In: The Cochrane Collaboration, editor. Cochrane Database of Systematic Reviews. Chichester, UK: John Wiley & Sons, Ltd; 2009. http://doi.wiley.com/10.1002/14651858. CD007483.pub2
- Thwing J, Eisele TP, Steketee RW. Protective efficacy of malaria case management and intermittent preventive treatment for preventing malaria mortality in children: a systematic review for the Lives Saved Tool. BMC Public Health. 2011; 11: S14. <u>https://doi.org/10.1186/1471-2458-11-S3-S14</u> PMID: 21501431
- World Health Organization. Guidelines for the treatment of malaria. Geneva, Switzerland: World Health Organization; 2015.
- D'Acremont V, Kilowoko M, Kyungu E, Philipina S, Sangu W, Kahama-Maro J, et al. Beyond Malaria— Causes of Fever in Outpatient Tanzanian Children. N Engl J Med. 2014; 370: 809–817. https://doi.org/ 10.1056/NEJMoa1214482 PMID: 24571753
- 8. World Health Organization. World malaria report 2018. Geneva, Switzerland: World Health Organization; 2018.
- Thiam S, Thior M, Faye B, Ndiop M, Diouf ML, Diouf MB, et al. Major Reduction in Anti-Malarial Drug Consumption in Senegal after Nation-Wide Introduction of Malaria Rapid Diagnostic Tests. Pied S, editor. PLoS ONE. 2011; 6: e18419. https://doi.org/10.1371/journal.pone.0018419 PMID: 21494674

- Altaras R. Integrating mRDTs into the health system in Uganda: Preparing health workers for routine use of malaria rapid diagnostic tests [Internet]. London, UK: Malaria Consortium; 2014 [cited 2020 Apr 9]. https://www.malariaconsortium.org/resources/publications/323/integrating-mrdts-into-the-healthsystem-in-uganda-preparing-health-workers-for-routine-use-of-malaria-rapid-diagnostic-tests
- President's Malaria Initiative. Community Health Workers Empowered to use RDTs in Rwanda [Internet]. 2013 [cited 2020 Apr 9]. https://www.pmi.gov/news/stories-from-the-field/stories-from-the-fielddetail/community-health-workers-empowered-to-use-rdts-in-rwanda
- Smith Gueye C, Gerigk M, Newby G, Lourenco C, Uusiku P, Liu J. Namibia's path toward malaria elimination: a case study of malaria strategies and costs along the northern border. BMC Public Health. 2014; 14: 1190. https://doi.org/10.1186/1471-2458-14-1190 PMID: 25409682
- Poyer S, Musuva A, Njoki N, Okara R, Cutherell A, Sievers D, et al. Fever case management at private health facilities and private pharmacies on the Kenyan coast: analysis of data from two rounds of client exit interviews and mystery client visits. Malar J. 2018; 17: 112. https://doi.org/10.1186/s12936-018-2267-8 PMID: 29534750
- Namuyinga RJ, Mwandama D, Moyo D, Gumbo A, Troell P, Kobayashi M, et al. Health worker adherence to malaria treatment guidelines at outpatient health facilities in southern Malawi following implementation of universal access to diagnostic testing. Malar J. 2017; 16: 40. https://doi.org/10.1186/ s12936-017-1693-3 PMID: 28114942
- Taffese HS, Hemming-Schroeder E, Koepfli C, Tesfaye G, Lee M, Kazura J, et al. Malaria epidemiology and interventions in Ethiopia from 2001 to 2016. Infect Dis Poverty. 2018; 7: 103. https://doi.org/10. 1186/s40249-018-0487-3 PMID: 30392470
- Masanja IM, Selemani M, Amuri B, Kajungu D, Khatib R, Kachur SP, et al. Increased use of malaria rapid diagnostic tests improves targeting of anti-malarial treatment in rural Tanzania: implications for nationwide rollout of malaria rapid diagnostic tests. Malar J. 2012; 11: 221. https://doi.org/10.1186/ 1475-2875-11-221 PMID: 22747655
- Lechthaler F, Matthys B, Lechthaler-Felber G, Likwela JL, Mavoko HM, Rika JM, et al. Trends in reported malaria cases and the effects of malaria control in the Democratic Republic of the Congo. PLoS ONE. 2019; 14: e0219853. https://doi.org/10.1371/journal.pone.0219853 PMID: 31344062
- **18.** World Health Organization, editor. Guidelines for the treatment of malaria. Geneva, Switzerland: World Health Organization; 2006.
- **19.** World Health Organization. Guidelines for the treatment of malaria. 2nd ed. Geneva, Switzerland: World Health Organization; 2010.
- Zhao J, Lama M, Korenromp E, Aylward P, Shargie E, Filler S, et al. Adoption of Rapid Diagnostic Tests for the Diagnosis of Malaria, a Preliminary Analysis of the Global Fund Program Data, 2005 to 2010. Noor AM, editor. PLoS ONE. 2012; 7: e43549. https://doi.org/10.1371/journal.pone.0043549 PMID: 22952703
- Kenangalem E, Poespoprodjo JR, Douglas NM, Burdam FH, Gdeumana K, Chalfein F, et al. Malaria morbidity and mortality following introduction of a universal policy of artemisinin-based treatment for malaria in Papua, Indonesia: A longitudinal surveillance study. PLoS Med. 2019; 16: e1002815. <u>https:// doi.org/10.1371/journal.pmed.1002815</u> PMID: 31167228
- Macarayan E, Papanicolas I, Jha A. The quality of malaria care in 25 low-income and middle-income countries. BMJ Glob Health. 2020; 5: e002023. https://doi.org/10.1136/bmjgh-2019-002023 PMID: 32133188
- 23. Johansson EW, Gething PW, Hildenwall H, Mappin B, Petzold M, Peterson SS, et al. Diagnostic Testing of Pediatric Fevers: Meta-Analysis of 13 National Surveys Assessing Influences of Malaria Endemicity and Source of Care on Test Uptake for Febrile Children under Five Years. Yukich J, editor. PLoS ONE. 2014; 9: e95483. https://doi.org/10.1371/journal.pone.0095483 PMID: 24748201
- O'Boyle S, Bruxvoort KJ, Ansah EK, Burchett HED, Chandler CIR, Clarke SE, et al. Patients with positive malaria tests not given artemisinin-based combination therapies: a research synthesis describing under-prescription of antimalarial medicines in Africa. BMC Med. 2020; 18: 17. <a href="https://doi.org/10.1186/s12916-019-1483-6">https://doi.org/10.1186/s12916-019-1483-6</a> PMID: 31996199
- 25. Bennett A, Bisanzio D, Yukich JO, Mappin B, Fergus CA, Lynch M, et al. Population coverage of artemisinin-based combination treatment in children younger than 5 years with fever and Plasmodium falciparum infection in Africa, 2003–2015: a modelling study using data from national surveys. Lancet Glob Health. 2017; 5: e418–e427. https://doi.org/10.1016/S2214-109X(17)30076-1 PMID: 28288746
- Alonso PL, Tanner M. Public health challenges and prospects for malaria control and elimination. Nat Med. 2013; 19: 150–155. https://doi.org/10.1038/nm.3077 PMID: 23389615
- 27. ACTwatch Group, Tougher S, Hanson K, Goodman C. What happened to anti-malarial markets after the Affordable Medicines Facility-malaria pilot? Trends in ACT availability, price and market share from

five African countries under continuation of the private sector co-payment mechanism. Malar J. 2017; 16: 173. https://doi.org/10.1186/s12936-017-1814-z PMID: 28441956

- World Health Organization, Global Malaria Programme. World malaria report 2017. Geneva, Switzerland: World Health Organization; 2017.
- Alegana VA, Maina J, Ouma PO, Macharia PM, Wright J, Atkinson PM, et al. National and sub-national variation in patterns of febrile case management in sub-Saharan Africa. Nat Commun. 2018; 9: 4994. https://doi.org/10.1038/s41467-018-07536-9 PMID: 30478314
- Opondo C, Allen E, Todd J, English M. Association of the Paediatric Admission Quality of Care score with mortality in Kenyan hospitals: a validation study. Lancet Glob Health. 2018; 6: e203–e210. https:// doi.org/10.1016/S2214-109X(17)30484-9 PMID: 29389541
- Muller O, Traore C, Becher H, Kouyate B. Malaria morbidity, treatment-seeking behaviour, and mortality in a cohort of young children in rural Burkina Faso. Trop Med Int Health. 2003; 8: 290–296. <u>https://doi.org/10.1046/j.1365-3156.2003.01030.x PMID: 12667146</u>
- 32. Biai S, Rodrigues A, Gomes M, Ribeiro I, Sodemann M, Alves F, et al. Reduced in-hospital mortality after improved management of children under 5 years admitted to hospital with malaria: randomised trial. BMJ. 2007; 335: 862. https://doi.org/10.1136/bmj.39345.467813.80 PMID: 17954513
- Kruk ME, Gage AD, Joseph NT, Danaei G, García-Saisó S, Salomon JA. Mortality due to low-quality health systems in the universal health coverage era: a systematic analysis of amenable deaths in 137 countries. The Lancet. 2018; 392(10160): 2203–2212. <u>https://doi.org/10.1016/S0140-6736(18)31668-4</u> PMID: 30195398
- Kabaghe AN, Visser BJ, Spijker R, Phiri KS, Grobusch MP, van Vugt M. Health workers' compliance to rapid diagnostic tests (RDTs) to guide malaria treatment: a systematic review and meta-analysis. Malar J. 2016; 15: 163. https://doi.org/10.1186/s12936-016-1218-5 PMID: 26979286
- Boyce MR, O'Meara WP. Use of malaria RDTs in various health contexts across sub-Saharan Africa: a systematic review. BMC Public Health. 2017; 17: 470. https://doi.org/10.1186/s12889-017-4398-1 PMID: 28521798
- Bruxvoort KJ, Leurent B, Chandler CIR, Ansah EK, Baiden F, Björkman A, et al. The Impact of Introducing Malaria Rapid Diagnostic Tests on Fever Case Management: A Synthesis of Ten Studies from the ACT Consortium. Am J Trop Med Hyg. 2017; 97: 1170–1179. <u>https://doi.org/10.4269/ajtmh.16-0955</u> PMID: 28820705
- Martin T, Eliades MJ, Wun J, Burnett SM, Alombah F, Ntumy R, et al. Effect of Supportive Supervision on Competency of Febrile Clinical Case Management in Sub-Saharan Africa. Am J Trop Med Hyg. 2019; 100: 882–888. https://doi.org/10.4269/ajtmh.18-0365 PMID: 30793696
- ICF. The DHS Program: SPA Overview [Internet]. 2017 [cited 2019 Feb 7]. https://dhsprogram.com/ What-We-Do/Survey-Types/SPA.cfm
- 39. The DHS Program User Forum: Service Provision Assessment (SPA). » How was data collected in bigger hospitals in the Malawi SPA inventory questionnaire 2013? [Internet]. 2017 [cited 2019 Dec 5]. https://userforum.dhsprogram.com/index.php?t=msg&th=6578&start=0&
- 40. Taylor C, Linn A, Wang W, Moussa H. Gaps in Service Utilization and Service Provision: An analysis of DHS and SPA Malaria Data from Malawi, Senegal, and Tanzania. Rockville, Maryland: ICF; 2018. Report No.: 67.
- World Health Organization. Emergence and spread of artemisinin resistance calls for intensified efforts to withdraw oral artemisinin-based monotherapy from the market [Internet]. Report No: WHO/HTM/ GMP/2014.3. 2014 [cited 2019 Feb 7]. http://www.who.int/malaria/publications/atoz/oral-artemisininbased-monotherapies-1may2014.pdf
- Colin Cameron A, Miller DL. A Practitioner's Guide to Cluster-Robust Inference. J Hum Resour. 2015; 50: 317–372. https://doi.org/10.3368/jhr.50.2.317
- 43. StataCorp. Stata Statistical Software: Release 14. College Station, TX: STataCorp LP; 2015.
- D'Acremont V, Lengeler C, Mshinda H, Mtasiwa D, Tanner M, Genton B. Time To Move from Presumptive Malaria Treatment to Laboratory-Confirmed Diagnosis and Treatment in African Children with Fever. PLoS Med. 2009; 6: e252. https://doi.org/10.1371/journal.pmed.0050252 PMID: 19127974
- **45.** Drakeley C, Reyburn H. Out with the old, in with the new: the utility of rapid diagnostic tests for malaria diagnosis in Africa. Trans R Soc Trop Med Hyg. 2009; 103: 333–337. <u>https://doi.org/10.1016/j.trstmh.</u> 2008.10.003 PMID: 19019399
- 46. Dondorp AM, Fanello CI, Hendriksen IC, Gomes E, Seni A, Chhaganlal KD, et al. Artesunate versus quinine in the treatment of severe falciparum malaria in African children (AQUAMAT): an open-label, randomised trial. The Lancet. 2010; 376: 1647–1657. https://doi.org/10.1016/S0140-6736(10)61924-1 PMID: 21062666

- Ford NP, de Smet M, Kolappa K, White NJ. Responding to the evidence for the management of severe malaria: Editorial. Trop Med Int Health. 2011; 16: 1085–1086. https://doi.org/10.1111/j.1365-3156. 2011.02810.x PMID: 21689220
- Wongsrichanalai C, Sibley CH. Fighting drug-resistant Plasmodium falciparum: the challenge of artemisinin resistance. Clin Microbiol Infect. 2013; 19: 908–916. <u>https://doi.org/10.1111/1469-0691.12316</u> PMID: 24033691
- 49. Burchett HED, Leurent B, Baiden F, Baltzell K, Björkman A, Bruxvoort K, et al. Improving prescribing practices with rapid diagnostic tests (RDTs): synthesis of 10 studies to explore reasons for variation in malaria RDT uptake and adherence. BMJ Open. 2017; 7: e012973. <u>https://doi.org/10.1136/bmjopen-2016-012973 PMID: 28274962</u>
- Zurovac D, Tibenderana JK, Nankabirwa J, Ssekitooleko J, Njogu JN, Rwakimari JB, et al. Malaria case-management under artemether-lumefantrine treatment policy in Uganda. Malar J. 2008; 7: 181. https://doi.org/10.1186/1475-2875-7-181 PMID: 18803833
- 51. Mishra N, Gupta R, Singh S, Rana R, Shahi B, Das MK, et al. Insights following change in drug policy: a descriptive study for antimalarial prescription practices in children of public sector health facilities in Jharkhand state of India. J Vector Borne Dis. 2013; 50: 271–277. PMID: 24499849
- Hennessee I, Chinkhumba J, Briggs-Hagen M, Bauleni A, Shah MP, Chalira A, et al. Household costs among patients hospitalized with malaria: evidence from a national survey in Malawi, 2012. Malar J. 2017; 16: 395. https://doi.org/10.1186/s12936-017-2038-y PMID: 28969643
- 53. Opwora A, Waweru E, Toda M, Noor A, Edwards T, Fegan G, et al. Implementation of patient charges at primary care facilities in Kenya: implications of low adherence to user fee policy for users and facility revenue. Health Policy Plan. 2015; 30: 508–517. https://doi.org/10.1093/heapol/czu026 PMID: 24837638
- Mikkelsen-Lopez I, Shango W, Barrington J, Ziegler R, Smith T, deSavigny D. The challenge to avoid anti-malarial medicine stock-outs in an era of funding partners: the case of Tanzania. Malar J. 2014; 13: 181. https://doi.org/10.1186/1475-2875-13-181 PMID: 24885420
- Zurovac D. Predictors of the quality of health worker treatment practices for uncomplicated malaria at government health facilities in Kenya. Int J Epidemiol. 2004; 33: 1080–1091. https://doi.org/10.1093/ije/ dyh253 PMID: 15256523
- 56. Skarbinski J, Ouma PO, Causer LM, Kariuki SK, Barnwell JW, Alaii JA, et al. Effect of malaria rapid diagnostic tests on the management of uncomplicated malaria with artemether-lumefantrine in Kenya: a cluster randomized trial. Am J Trop Med Hyg. 2009; 80: 919–926. PMID: 19478249
- Cundill B, Mbakilwa H, Chandler CI, Mtove G, Mtei F, Willetts A, et al. Prescriber and patient-oriented behavioural interventions to improve use of malaria rapid diagnostic tests in Tanzania: facility-based cluster randomised trial. BMC Med. 2015; 13: 118. https://doi.org/10.1186/s12916-015-0346-z PMID: 25980737
- Mbacham WF, Mangham-Jefferies L, Cundill B, Achonduh OA, Chandler CI, Ambebila JN, et al. Basic or enhanced clinician training to improve adherence to malaria treatment guidelines: a cluster-randomised trial in two areas of Cameroon. Lancet Glob Health. 2014; 2: e346–e358. https://doi.org/10.1016/ S2214-109X(14)70201-3 PMID: 25103303
- Leslie HH, Gage A, Nsona H, Hirschhorn LR, Kruk ME. Training And Supervision Did Not Meaningfully Improve Quality Of Care For Pregnant Women Or Sick Children In Sub-Saharan Africa. Health Aff (Millwood). 2016; 35: 1716–1724. https://doi.org/10.1377/hlthaff.2016.0261 PMID: 27605655
- Rowe AK, Rowe SY, Peters DH, Holloway KA, Chalker J, Ross-Degnan D. Effectiveness of strategies to improve health-care provider practices in low-income and middle-income countries: a systematic review. Lancet Glob Health. 2018; 6: e1163–e1175. https://doi.org/10.1016/S2214-109X(18)30398-X PMID: 30309799
- Kruk ME, Gage AD, Arsenault C, Jordan K, Leslie HH, Roder-DeWan S, et al. High-quality health systems in the Sustainable Development Goals era: time for a revolution. Lancet Glob Health. 2018; 6: e1196–e1252. https://doi.org/10.1016/S2214-109X(18)30386-3 PMID: 30196093
- Frenk J, Chen L, Bhutta ZA, Cohen J, Crisp N, Evans T, et al. Health professionals for a new century: transforming education to strengthen health systems in an interdependent world. The Lancet. 2010; 376: 1923–1958. https://doi.org/10.1016/S0140-6736(10)61854-5
- Bowser D, Abbas Y, Odunleye T, Broughton E, Bossert T. Pilot study of quality of care training and knowledge in Sub-Saharan African medical schools. Int J Med Educ. 2017; 8: 276–282. https://doi.org/ 10.5116/ijme.595b.b38c PMID: 28753130
- Lange S, Mwisongo A, Mæstad O. Why don't clinicians adhere more consistently to guidelines for the Integrated Management of Childhood Illness (IMCI)? Soc Sci Med. 2014; 104: 56–63. https://doi.org/ 10.1016/j.socscimed.2013.12.020 PMID: 24581062

- 65. Shah JA, Emina JBO, Eckert E, Ye Y. Prompt access to effective malaria treatment among children under five in sub-Saharan Africa: a multi-country analysis of national household survey data. Malar J. 2015; 14: 329. https://doi.org/10.1186/s12936-015-0844-7 PMID: 26303581
- 66. Eisele TP, Silumbe K, Yukich J, Hamainza B, Keating J, Bennett A, et al. Measuring Coverage in MNCH: Accuracy of Measuring Diagnosis and Treatment of Childhood Malaria from Household Surveys in Zambia. PLoS Med. 2013; 10: e1001417. <u>https://doi.org/10.1371/journal.pmed.1001417</u> PMID: 23667337
- Ashton RA, Doumbia B, Diallo D, Druetz T, Florey L, Taylor C, et al. Measuring malaria diagnosis and treatment coverage in population-based surveys: a recall validation study in Mali among caregivers of febrile children under 5 years. Malar J. 2019; 18(1): 3. https://doi.org/10.1186/s12936-018-2636-3 PMID: 30602376
- Steinhardt LC, Chinkhumba J, Wolkon A, Luka M, Luhanga M, Sande J, et al. Quality of Malaria Case Management in Malawi: Results from a Nationally Representative Health Facility Survey. Eisele T, editor. PLoS ONE. 2014; 9: e89050. https://doi.org/10.1371/journal.pone.0089050 PMID: 24586497
- 69. Shah MP, Briggs-Hagen M, Chinkhumba J, Bauleni A, Chalira A, Moyo D, et al. Adherence to national guidelines for the diagnosis and management of severe malaria: a nationwide, cross-sectional survey in Malawi, 2012. Malar J. 2016; 15: 369. https://doi.org/10.1186/s12936-016-1423-2 PMID: 27430311
- 70. Sears D, Kigozi R, Mpimbaza A, Kakeeto S, Sserwanga A, Staedke SG, et al. Anti-malarial prescription practices among outpatients with laboratory-confirmed malaria in the setting of a health facility-based sentinel site surveillance system in Uganda. Malar J. 2013; 12: 252. https://doi.org/10.1186/1475-2875-12-252 PMID: 23870515
- Zurovac D, Githinji S, Memusi D, Kigen S, Machini B, Muturi A, et al. Major Improvements in the Quality of Malaria Case-Management under the "Test and Treat" Policy in Kenya. Carvalho LH, editor. PLoS ONE. 2014; 9: e92782. https://doi.org/10.1371/journal.pone.0092782 PMID: 24663961
- 72. Bruxvoort K, Kalolella A, Nchimbi H, Festo C, Taylor M, Thomson R, et al. Getting antimalarials on target: impact of national roll-out of malaria rapid diagnostic tests on health facility treatment in three regions of Tanzania. Trop Med Int Health. 2013; 18: 1269–1282. https://doi.org/10.1111/tmi.12168 PMID: 23937722
- 73. Mubi M, Kakoko D, Ngasala B, Premji Z, Peterson S, Björkman A, et al. Malaria diagnosis and treatment practices following introduction of rapid diagnostic tests in Kibaha District, Coast Region, Tanzania. Malar J. 2013; 12: 293. https://doi.org/10.1186/1475-2875-12-293 PMID: 23977904
- 74. Bonful HA, Awua AK, Adjuik M, Tsekpetse D, Adanu RMK, Nortey PA, et al. Extent of inappropriate prescription of artemisinin and anti-malarial injections to febrile outpatients, a cross-sectional analytic survey in the Greater Accra region, Ghana. Malar J. 2019; 18(1): 331. https://doi.org/10.1186/s12936-019-2967-8 PMID: 31558149
- 75. Onwujekwe O, Hanson K, Uzochukwu B. Do poor people use poor quality providers? Evidence from the treatment of presumptive malaria in Nigeria. Trop Med Int Health. 2011; 16: 1087–1098. <u>https://doi.org/10.1111/j.1365-3156.2011.02821.x PMID: 21702870</u>
- 76. Odaga J, Sinclair D, Lokong JA, Donegan S, Hopkins H, Garner P. Rapid diagnostic tests versus clinical diagnosis for managing people with fever in malaria endemic settings. In: The Cochrane Collaboration, editor. Cochrane Database of Systematic Reviews. Chichester, UK: John Wiley & Sons, Ltd; 2014. http://doi.wiley.com/10.1002/14651858.CD008998.pub2
- Leonard K, Masatu MC. Outpatient process quality evaluation and the Hawthorne Effect. Soc Sci Med. 2006; 63: 2330–2340. https://doi.org/10.1016/j.socscimed.2006.06.003 PMID: 16887245
- Leurent B, Reyburn H, Muro F, Mbakilwa H, Schellenberg D. Monitoring patient care through health facility exit interviews: an assessment of the Hawthorne effect in a trial of adherence to malaria treatment guidelines in Tanzania. BMC Infect Dis. 2016; 16: 59. https://doi.org/10.1186/s12879-016-1362-0 PMID: 26842751
- 79. World Health Organization, Roll Back Malaria Partnership. High burden to high impact: a targeted malaria response [Internet]. 2018 [cited 2019 Dec 5]. https://www.who.int/malaria/publications/atoz/high-impact-response/en/
- Rowe AK. Potential of integrated continuous surveys and quality management to support monitoring, evaluation, and the scale-up of health interventions in developing countries. Am J Trop Med Hyg. 2009; 80: 971–979. PMID: 19478260
- Alonso P, Noor AM. The global fight against malaria is at crossroads. The Lancet. 2017; 390: 2532– 2534. https://doi.org/10.1016/S0140-6736(17)33080-5